
% -----------------------------*- LaTeX -*------------------------------------
\documentclass[12pt]{report}
\usepackage{scribe_15853_f19}
\usepackage{graphicx}
\begin{document}

%%TODO: Please update the following three fields for each scribe notes
\scribe{Rashmi Vinayak}		% required
\lecturenumber{1}			% required, must be a number
\lecturedate{September 3 2019}		% required, omit year

\maketitle

% ----------------------------------------------------------------------------

\section{Recap From Last Class}
This was the first lecture of this course!

\section{Overview}
In this lecture, we saw an overview of the course, logistics and started learning about error correcting codes. 

\section{Instructions for scribing}
\begin{itemize}
	\item Scribe notes should be clear and detailed, using figures whenever necessary.
	\item Use macros for notation. If you not used Latex macros, read and learn about them! Latex macros are very powerful and are essential for efficient technical writing in Latex.
	\item Use theorem, proposition, lemma and similar other environments whenever necessary as presented in class.
\end{itemize}



\end{document}
